Author’s response to reviews

Title: Changes in Frailty among Community-Dwelling Older Chinese Adults and its Predictors: Evidence from a Two-Year Longitudinal Study

Authors:

Bo Ye (17111020016@fudan.edu.cn)
Hao Chen (haochen18@fudan.edu.cn)
Limei Huang (hlm950526@163.com)
Ye Ruan (ruanye@scdc.sh.cn)
Shige Qi (qishige@ncncd.chinacdc.cn)
Yanfei Guo (guoyanfei@scdc.sh.cn)
Zhezhou Huang (huangzhezhou@scdc.sh.cn)
Shuangyuan Sun (sunshuangyuan@scdc.sh.cn)
Xiuqin Chen (xqchen2013@163.com)
Yan Shi (Shiyan@scdc.sh.cn)
Junling Gao (jlgao@fudan.edu.cn)
Yonggen Jiang (Sjjkzx1106@126.com)

Version: 2 Date: 16 Mar 2020

Author’s response to reviews:

Editor Comments:

Page 2: Line 21-23: During the next 2 years, there were 127 deaths and 23 participants lost to follow-up. Of those who were lost to follow-up, were unable to recontact them and some may have emigrated to another city. Five of these participants had prefrail and 18 had robust status.

Response to Editor Comments: Thank you for your grammatical errors revision. We had adopted all suggestions.
Ivan Aprahamian (Reviewer 1): No further comments. Thanks.

Response to Reviewer 1: Thank you for your suggestion.

Zhijun Bao, Ph.D. (Reviewer 2): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format. Please overwrite this text when adding your comments to the authors.

Response to Reviewer 2: We think these not comments to the manuscript.